Author's response to reviews

**Title:** Regional differences in mortality in Greece (1984-2004): the case of Thrace

**Authors:**

Panagiotis PP Papastergiou (panpapast@med.uth.gr)  
George GR Rachiotis (gsrachmed@yahoo.com)  
Konstantina KP Polyzou (panpapast@med.uth.gr)  
Christos CZ Zilidis (zilidis@gmail.com)  
Christos CC Hadjichristodoulou (xhatzi@med.uth.gr)

**Version:** 5  **Date:** 8 August 2008

**Author's response to reviews:** see over
BMC Public Health

Larissa, August 6, 2008

Dear Editor,

Thank you and the reviewer for constructive comments helping us to improve further our manuscript.

Please find attached the final form of the manuscript. We have revised the text in order to avoid direct replication of the reviewer’s comments.

Yours sincerely,

Christos Chadjichristodoulou
Ass. Professor of Epidemiology
Department of Hygiene and Epidemiology
School of Medicine
University of Thessaly
e-mail: xhatzi@med.uth.gr